Author's response to reviews

Title: Prevalence and risk factors of cervical neoplasia: a cervical cancer screening program in Beijing

Authors:

Lixin Tao (taolixin.2008@163.com)
Lili Han (hanll2004@126.com)
Xia Li (lixia_new@163.com)
Qi Gao (gaoqi@ccmu.edu.cn)
Lei Pan (panlei_helen@163.com)
Lijuan Wu (wujuan811017@163.com)
Yanxia Luo (lyx100@ccmu.edu.cn)
Wei Wang (wei.wang@ecu.edu.au)
Zihe Zheng (myszihe@126.com)
Xiuhua Guo (guoxiu@ccmu.edu.cn)

Version: 11 Date: 1 November 2014

Author's response to reviews: see over
MS: 9618051549638581

Title: Prevalence and risk factors of cervical neoplasia: a cervical cancer screening program in Beijing

Referee 1:

The specific responses to Referee’s comments

Comment 1) I believe that English needs to be corrected

Response:

The paper has been copyedited by a native-English speaker with scientific expertise to improve the style of written English as suggested. The tables and references have also been formatted to conform to the journal style.

Comment 2) More current references (2013-2014) should be inserted.

Response:

Based on the latest literature about cervical cancer and its related factors, I have inserted 11 current references published in 2013 or 2014 in the revised manuscript and deleted 7 old references from the manuscript.

The new references inserted in the revised manuscript are as follows:


The old references deleted from the manuscript are as follows:


Additional Editorial Request

The specific response to Editor’s comment

Comment 1) We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service.

Response:

The paper has been copyedited by a native-English speaker with scientific expertise from Edanz to improve the style of written English as suggested. The tables and references have also been formatted to conform to the journal style.